Author's response to reviews

Title: Multidisciplinary Collaborative Care for Depressive Disorder in the Occupational Health Setting: design of a randomized controlled trial and cost-effectiveness study.

Authors:

Moniek C Vlasveld (mvlasveld@trimbos.nl)
Johannes R Anema (h.anema@vumc.nl)
Aartjan TF Beekman (aartjanb@ggzba.nl)
Willems van Mechelen (w.vanmechelen@vumc.nl)
Rob Hoedeman (rob.hoedeman@arboned.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Frans F Rutten (f.rutten@erasmusmc.nl)
Leona Hakkaart-van Roijen (hakkaart@bm.eur.nl)
Christina van der Feltz-Cornelis (cfeltz@trimbos.nl)

Version: 5 Date: 29 April 2008

Author's response to reviews:

Because we had made some changes to the design recently, I adjusted the manuscript. We removed some questionnaires from the measurements and adjusted one exclusion criterion.